Author's response to reviews

Title: Mothers' experience of their contact with their stillborn infant: An interpretative phenomenological analysis.

Authors:

Kirsty Ryninks (Kirsty.Ryninks@UHBristol.nhs.uk)
Cara Roberts-Collins (c.roberts-collins@bath.ac.uk)
Kirstie McKenzie-Mcharg (kirstie.mckenzie-mcharg@swft.nhs.uk)
Antje Horsch (antje.horsch@chuv.ch)

Version: 2 Date: 13 January 2014

Author's response to reviews:

We have named the institutional committees that provided ethical approval for the study (p. 6), as requested in letter received on 13.01.2014.